Author’s response to reviews

Title: Safety of a bivalent, killed, whole-cell oral cholera vaccine in pregnant women in Bangladesh: Evidence from a randomized placebo-controlled trial

Authors:

Firdausi Qadri (fqadri@icddrb.org)
Ashraful Khan (ashrafk@icddrb.org)
Md Ali (mali25@jhu.edu)
Alamgir Kabir (alamgir.kabir@icddrb.org)
Arifuzzaman Khan (arifkhan@icddrb.org)
Md Taufiqul Islam (taufiqulislam@icddrb.org)
Afroza Akter (afrozaakter@icddrb.org)
Fahima Chowdhury (fchowdhury@icddrb.org)
Amit Saha (amit@icddrb.org)
Iqbal Khan (iqbalansary@gmail.com)
Nirod Saha (nirod@icddrb.org)
John Clemens (jcemens@icddrb.org)

Version: 1 Date: 05 Mar 2018

Author’s response to reviews:

Response to Editor's comments:

Comment 1: Completed CONSORT checklist has been attached.

Comment 2: Consenting process for minor participant has been included in the method section (Page: 7; Line: 137).

Comment 3: Trial registration date has been included alongside the trial registration number (Page: 14; Line: 311).

Comment 4: Rewording has been done in the manuscript (Page: 11, 12; Line: 249, 250 and 251)